Author's response to reviews

Title: Use of aspirin for primary and secondary prevention of cardiovascular disease in diabetic patients in an ambulatory care setting in Spain

Authors:

Antoni Sicras-Mainar (asicras@bsa.gs)
Ruth Navarro-Artieda (rnavarro.germanstrias@gencat.net)
Javier Rejas-Gutierrez (javier.rejas@pfizer.com)
Jaime Fernandez-de-Bobadilla (jaime.fernandez@pfizer.com)
Xavier Frias-Garrido (xfrias@bsa.gs)
Rafael Ruiz-Ruiz (asicras@terra.es)

Version: 7 Date: 14 October 2007

Author’s response to reviews: see over

We really appreciate and thank you the comments and suggestions made on this manuscript. AS you can see, the format of manuscript has been reviewed and modified accordingly with journal guidelines.

The BioMed Central Editorial Production Team
Formatting changes requested

1. Lines- Please remove all visible lack lines from the title page
   Modified in the text

2. Please remove 'Page title:' from the title page.
   Modified in the text

3. Please ensure that the authors are listed in the same order in the submission system and the manuscript.
   Modified in the text

4. Please do not place each author on a new line.
   Modified in the text

5. Please remove the number tables/number figures/ running title from the title page.
   Modified in the text

6. Please remove 'Notifications of Conflicts of Interest and Ethical Adherence' from the title page- you have stated this in the competing interests section.
   Modified in the text

7. Please delete the key words from the abstract.
   Modified in the text

8. References - The reference list should contain all authors titles, if there are 30 or below (ref 5, 8, 15). The term "et al." should not be used.
   Modified in the text (ref 5, 8, 15)

9. Please cite the references in order- we notice that the references jump from 28 to 30.
   Modified in the text

10. Table cells - Tables must be divided into cells/fields. The borders of tables included as part of the manuscript file must be visible black lines. Tables generated with tabbed text are not acceptable. Please ensure that text alignment in columns is clear and consistent. We are unable to correctly display merged cells where the merged cell crosses rows.
    Modified in the manuscript - tables